Author's response to reviews

Title: The association between dental caries and television viewing among Chinese adolescents in Guangxi, China.

Authors:

  Xiaojuan Zeng (xiaojuan.zeng@gmail.com)
  Aubrey Sheiham (a.sheiham@ucl.ac.uk)
  Wael Sabbah (wael.sabbah@kcl.ac.uk)

Version: 4 Date: 8 November 2014

Author's response to reviews: see over
Re: MS MS: 6400844611367952. The association between dental caries and television viewing among Chinese adolescents in Guangxi, China

We would like to thank the editor and the reviewers for their constructive critique of the manuscript. We have now addressed all additional comments from the editor and amended the manuscript.

The following changes have been made

Methods: page 4, lines 71-74

The Department of Education of Guangxi and Guangxi Province Center for Disease Control and Prevention gave permission to use data from the 2010 National Physical Fitness and Health Surveillance Survey.

Acknowledgement: page 10 lines 209-211

The 2010 National Physical Fitness and Health Surveillance survey was funded by the Department of Finance of Guangxi. This study was not funded, none of the authors received any fund to participate in this study.